Author's response to reviews

Title: Domestic Violence against Women in Eastern Sudan.

Authors:

AbdelAziem A Ali (abuzianab73@yahoo.com)
Khalid Yasin (khalidym@hotmail.com)
Rawia Omer (widaltestforbrucella2013@gmail.com)

Version: 5  Date: 10 October 2014

Author's response to reviews:

1- We specifically named which “validated and widely used questionnaire” that we have used and we provided a reference to a study where it was validated, please see the methodology and the references.

All these changes have been highlighted to be traced easily.

Dr. Ali